Author’s response to reviews

Title: Feasibility and acceptability of combining cognitive behavioural therapy techniques with swallowing therapy in head and neck cancer dysphagia

Authors:

Jo Patterson (joanne.patterson@ncl.ac.uk)

Michaela Fay (mail@michaelafay.net)

Catherine Exley (catherine.exley@northumbria.ac.uk)

Elaine McColl (e.mccoll@ncl.ac.uk)

Matthew Breckons (Matthew.Breckons@newcastle.ac.uk)

Vincent Deary (vincent.deary@northumbria.ac.uk)

Version: 3 Date: 06 Nov 2017

Author’s response to reviews:

Many thanks for the opportunity to re-submit this manuscript. Regarding the revisions requested, these have been revised as follows

1- A signed form for change of authorship had been submitted to the editorial office.

2- Re quotations used from participants - to maintain confidentiality their age and subject number has been used in place of gender and age.

3- I have clarified that written consent was also 'informed' throughout the manuscript.

4- Specific Authors' Contributions for all of the authors have been included.